Author's response to reviews

Title: The e-Health Implementation Toolkit: Qualitative evaluation across four European countries

Authors:

Anne MacFarlane (anne.macfarlane@nuigalway.ie)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Elizabeth Murray (elizabeth.murray@pcps.ucl.ac.uk)
David F Heaney (d.heaney@abdn.ac.uk)
Mary Wakeling (mary@bongosoft.co.uk)
Ulla-Maija Pesola (umpesola@informatik.umu.se)
Eva Lindh Waterworth (eva@informatik.umu.se)
Frank Larsen (frank.larsen@telemed.no)
Minna Makiniemi (minna.makiniemi@ppshp.fi)
Ikka Winblad (ilkka.winblad@oulu.fi)

Version: 5 Date: 12 November 2011

Author's response to reviews:

November 12th 2011

Dear Editors,

I am now submitting a revised version of this manuscript.

I have changed David F Heaney to David J Heaney

Corrected Ikka to Ilkaa

Updated my own institutional affiliations to reflect my recent move from NUI Galway, Ireland to the University of Limerick, Ireland

I have removed Table 1 and incorporated the text into the main body of the manuscript - it is now indented as a means of distinguishing it from the remainder of the material in the main body.

With best wishes,

Anne MacFarlane, PhD
Professor in Primary Healthcare Research
Graduate Entry Medical School
University of Limerick
Limerick
Ireland